Author's response to reviews

Title: Oxidative stress markers are associated to vascular recurrence in non-cardioembolic stroke patients non-treated with statins

Authors:

David Brea (davidbrealopez@yahoo.es)
Jaume Roquer (JRoquer@imas.imim.es)
Joaquín Serena (jserenal@meditex.es)
Tomás Segura (tseguram@meditex.es)
José Castillo (jose.castillo@usc.es)

Version: 3 Date: 12 July 2012

Author's response to reviews: see over
Dear Editor of BMC Neurology,

Please find enclosed to this letter the revised version of our manuscript entitled "OXIDATIVE STRESS MARKERS ARE ASSOCIATED TO VASCULAR RECURRENCE IN NON-CARDIOEMBOLIC STROKE PATIENTS NON-TREATED WITH STATINS" and the response to reviewers document. An ethics statement have been included in the Methods section of the manuscript.

The material in this manuscript has not been previously published, and it has not been submitted to any other journal. I state, as corresponding author, that all the named authors have had an active part in the production of this work, and all have agreed to its publication. I take responsibility for the integrity of the data and the accuracy of the data analysis. I state that Dr. Roquer, Dr. Serena, Dr. Segura and Dr. Castillo report receiving consulting and advisory board fees from Bristol-Myers Squibb, however the company had nothing to do in result analysis or interpretation or in the decision to publish.

Yours sincerely,

José Castillo